ECONSTOR

Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bekiaris, Michalis; Markogiannopoulou, Antonia

Article

Value Relevance of General Government National Accounts with ESA2010 Accrual Accounting Framework. Association of ESA2010 Reporting Quality with Decision Making and Accounting Standardization

Journal of Accounting and Management Information Systems (JAMIS)

Provided in Cooperation with:

The Bucharest University of Economic Studies

Suggested Citation: Bekiaris, Michalis; Markogiannopoulou, Antonia (2022) : Value Relevance of General Government National Accounts with ESA2010 Accrual Accounting Framework. Association of ESA2010 Reporting Quality with Decision Making and Accounting Standardization, Journal of Accounting and Management Information Systems (JAMIS), ISSN 2559-6004, Bucharest University of Economic Studies, Bucharest, Vol. 21, Iss. 4, pp. 546-574, https://doi.org/10.24818/jamis.2022.04004

This Version is available at: https://hdl.handle.net/10419/310843

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



http://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Value relevance of general government national accounts with ESA2010 accrual accounting framework. Association of ESA2010 reporting quality with decision making and accounting standardisation

Michalis Bekiaris^a and Antonia Markogiannopoulou^{1a}

^a Department of Business Administration, School of Business, University of the Aegean, Chios, Greece

Abstract

Purpose: The necessity for a reliable set of international standards for the compilation of national accounts introduced the accrual accounting framework of the European System of Accounts (ESA 2010). Extensive efforts are made for convergence between Government Accounts (GA) and National Accounts (NA) under accounting standards.

Design/methodology/approach: This study investigated the quality of governmental NA, in the context of ESA 2010, from 1999 through 2019, using relative value relevance models of long-term government bond yields.

Findings: i) The accrual accounting framework ESA2010 generates value relevant (thus qualitative) NA financial reporting in EU member states and the United Kingdom (UK); ii) The financial variables that better interpret bonds' return, thus governments' necessity to borrow money, are showcased; iii) The usefulness of the ESA2010 conceptual framework for decision and policy making process considering that the accrual accounting basis fosters the financial reporting quality, is demonstrated; iv) The quality results set the premise for further discussion for the harmonization and alignment process of accounting standardisation with the ESA2010 accrual accounting framework as a resource for policy and decision making.

¹ Corresponding author: Antonia Markogiannopoulou, Department of Business Administration, School of Business, University of the Aegean, Chios, Greece, Tel. (+30) 6949972476, email addresses: a.markogiannopoulou@aegean.gr

Limitations: The methodology of value relevance models that is employed to assess the governmental statistical reporting quality with the ESA2010 accrual accounting framework is a private sector technique.

Originality: This study contributes theoretically as it fosters the quality of accrual accounting basis. It moreover provides an empirical and practical contribution by introducing relative value relevance econometric models that provide stakeholders with reliable information on the financial position and performance of the government. It showcases significant financial variables and coefficients of statistical reporting for each government. With its value relevant results, it supports evidence-based decision making, allows comparisons between EU governments and the UK, and contributes to increased transparency and accountability.

Keywords: IPSAS, EPSAS, ESA2010, national accounts, value relevance, bond return

JEL codes: H83, M41

1. Introduction

Public sector accrual-based accounting sets the breeding ground for the provision of high-quality statistics (European Commission (EC), 2012) at the European level for the harmonization of government accounting (GA) with financial statistical reporting. At the macroeconomic level, the United Nations System of National Accounts (SNA) 2008 draws the statistical perspective of financial reporting, whereas GA comprises the applicable micro-level point of view. National accounts (NA) have been produced and circulated with the support of the United Nations, the Organization for Economic Co-operation and Development (OECD), the International Monetary Fund (IMF), the World Bank Group, and the European Commission (EC). Since September 2014, in the European division, the corresponding variant is supported by the European System of Accounts (ESA) 2010.

National accounts are a set of macroeconomic financial records, allowing the comparison between countries while enabling decision making. The structure of ESA 2010 is in line with the global national accounting guidelines employed in the SNA2008. The Government Finance Statistics (GFS) information guidelines govern both SNA2008 and ESA2010 (González *et al.*, 2018). Statistical accounting is a comprehensive framework that provides a reliable and flexible macroeconomic chart of accounts for decision making and research purposes. In the European Union (EU), the significance of GA practices being in accordance with the NA, specifically ESA2010, is inescapable. These practices serve as the foundation upon which the EU's fiscal discipline and macroeconomic convergence requirements are evaluated (Jorge *et al.*, 2019).

Vol. 21, No. 4

This study primarily aims to highlight the contribution of the ESA accrual accounting framework to the value relevance (thus the quality) of statistical reporting of governmental NA. It aims to foster the analysis of accrual accounting harmonization processes in European division using the accounting frameworks of NA as key elements. In order to improve social, political, and economic decision making and accountability, it advocates value relevance tools that facilitate the examination and comparability of financial, budgetary, and aggregated statistical reports. With its findings, this study has practical implications as it i) concludes that ESA2010 generates value relevant (thus qualitative) NA financial reporting in EU member states and the United Kingdom (UK) on an accrual basis; ii) highlights the financial variables that better interpret bonds' return thus governments' necessity to borrow money iii) endorses the usefulness of the ESA2010 conceptual framework for the decision and policy making process considering that the accounting frameworks foster the financial reporting quality; iv) discusses the value relevance results in association with the harmonization and alignment process of accounting standardisation with the ESA2010 accrual accounting framework during the decision making process.

The remainder of this paper is organised as follows. Section 2 reviews the relevant literary and theoretical framework that sets the premise for the research questions developed in Section 3. Section 4 analyses the research methodology and sources. Section 5 provides the results of this research. Section 6 discusses the results in association with decision and policy making and accounting standardisation. Section 7 concludes with the contribution and the opportunities for accounting academics and practitioners.

2. Literary and theoretical framework

2.1 Value relevance in the public sector and public administration theory

According to international literature (Hung & Subramanyam, 2007) value relevance is defined as the ability of accounting figures to reflect the financial value of each legal entity. It can reflect accounting methods and accounting standards, auditing and legislation, and standard interpretation. Strong value relevance implies high quality accounting information (Ball & Brown, 1968; Ohlson, 1995, 2001). In the relative value relevance study, using the same value relevance model but with different accounting values corresponding to different standards or different periods or different groups, the R^2 ratio of the equations determines whether there is a significant value relevance difference and consequently qualitative variance in the accounting information (Lin & Chen, 2005).

Value relevance refers to the advantageous characteristics of financial statements to provide reliable and relevant accounting information as the primary criteria for

Vol. 21, No. 4

Value Relevance of General Government National Accounts with ESA2010 accrual accounting framework. Association of ESA2010 reporting quality with decision making and accounting standardization

enabling standard setters to choose between accounting practices and standard alternatives (Barth *et al.*, 2001). From a measurement and disclosure perspective, accounting information is relevant if there is a statistical association between financial information and prices or returns. This concept positions equity and debt holders as the focus of the definition. From Francis and Schipper's (1999) approach, value relevance influences investors' decision making as to whether or not to invest in companies' shares. So far, numerous specification models have varied in their treatment and perceptions of accounting practices and financial information depending on the valuation models used (Easton & Harris, 1991; Ohlson, 1995). Based on Kothari and Zimmerman (1995), even though price models are less biased, returns models are less problematic for econometric analysis and are commonly favoured. To associate accounting figures with returns, the accounting literature has used regression analysis of modified valuation models as a theoretical framework for this relationship.

Hung and Subramanyam (2007) defined value relevance as the ability of accounting to convey and reflect the fiscal value of legal entities, their corresponding accounting methods, and their accounting standards (auditing and legislation). Ball and Brown (1968), studying the relevance of earnings and shareholders' returns, concluded that the higher the value relevance is, the greater the quality of accounting information and therefore of earnings. Their research has established the basis for future studies to compare different accounting practices or standards. Easton and Harris (1991) contributed to the discussion by associating changes in earnings with returns, using a book valuation model to determine earnings levels' relationship with returns. Ohlson's (1995) valuation model is based on the principle that a firm's weighted value is linearly associated with owners' equity and earnings. Ohlson's model relates pricing to the major components of balance sheets and income statements. Francis and Schipper (1999) further investigated value relevance and correlated a firm's value with major components of equity and earnings. The value relevance of balance sheet measures is sensitive to the valuation principles applied to the various asset and debt components. Barth et al. (2001) claimed the balance sheet's distinctive roles were to provide information on liquidation values to facilitate loan decisions and monitor debt contracts.

We acknowledge certain limitations of the study. The value relevance of earnings and book value of equity on share prices and stock returns is a private sector technique to assess the quality of accounting information. The Generally Accepted Accounting Principles (GAAP) of the US and the International Financial Reporting Standards (IFRS) have long been competing for international acceptance by stock markets around the world as financial reporting standards. Subject cross-country comparisons of the value relevance of earnings and book values have been investigated between IFRS and US GAAP by Barth *et al.* (2012). Cross-country relative value relevance has also been studied during the IFRS convergence

Vol. 21, No. 4

(Srivastava & Muharam, 2021). Literature from a value relevance perspective has also focused on the transitioning period from local accounting standards to IFRS (Clarkson *et al.*, 2011; Kadri *et al.*, 2009; Tsalavoutas *et al.*, 2009). Evaluation of financial performance in terms of the value relevance of IFRS was considered significant not only for investors or stakeholders but also for those who wanted to invest at the international level (Temiz & Güleç, 2017). As accounting information is mostly related to debt markets (Ball & Shivakumar, 2008; Givoly *et al.*, 2017) this research deploys debt, equity, income and Euribor figures in association with bonds' return figures.

With the introduction and adoption of the accrual accounting system in the public sector as a result of New Public Management (NPM) reforms, which attempted to emulate the accounting practises and procedures used in the private sector, we initiate value relevance analysis in the public domain employing the accrual accounting framework ESA2010, to assess the quality of statistical reporting via bonds' return association. The NPM model represents new advances that were made to public administration theory as a new approach to public sector governance, with the aim to improve efficiency and accountability (Gomes *et al.*, 2015). The term itself was first introduced by academics to describe the techniques that developed during the 1980s as part of an effort to make public services more business-like and to improve their efficiency by using private sector management models and free market principles in the public sector (Lampropoulou & Oikonomou, 2018).

Accrual accounting, as the focal point of NPM reforms, provides significant information content to communicate financing decisions and allows the government to take a long-term view (Salato et al., 2022). Adoption related decisions seem to be more motivated by the objectives of accountability and transparency (Nitzl et al., 2020) to maintain the international harmonization process (Brito & Jorge, 2021). Developed countries, such as the UK, are leading this reform process and already have a high level of implementation of accrual accounting in the public sector (Ghani et al., 2019), in the belief that it will lead to better decision making (Hyndman & Connolly, 2011). Christiaens et al. (2015) underlined the connection of the International Public Sector Accounting Standards (IPSAS) standards with the general (NPM) approach. Subject reforms brought accrual accounting to public sector accounting (PSA) systems, in an approximation to business accounting, and the adaptation of the IPSAS framework to national standards. In EU member states, the PSA reforms seem to have been driven principally by external factors, such as EU guidelines (Directive 2011/85/UE) that mandate accrual accounting for fiscal reporting under the ESA 2010 and the Eurostat initiative to implement and develop the IPSAS-based European Public Sector Accounting Standards (EPSAS) (World Bank/ Public Sector Accounting and Reporting Program (PULSAR), 2021).

Vol. 21, No. 4

2.2 Conversion of GA to NA

The Excessive Deficit Procedure (EDP) Tables that each European country prepares, and Eurostat publishes for deficits and government debt, reflect the adjustments made for the ESA deficit from working balance thus the public accounting balance. Working balance is the only public figure that Eurostat publishes and is the broad national definition of the difference between revenue and expenditure. EDP tables record the variations between public budgetary accounting and NA for net lending/borrowing and permit the investigation of the discrepancies between the two systems through appropriate adjustments. These adjustments, made on a national level, are an indication of the deviation of public accounting standards from ESA 2010. Since this difference varies definitionally in each member state, statisticians get corresponding information from EDP inventories. Recognition criteria of the working balance (budgetary balance), whether cash, accrual, mixed or otherwise is also indicated (Dasí *et al.*, 2013; Eurostat, 2014).

Alignment of the GA practises and statistics is necessary to relegate the adjustments considerably when converting data from (GA) into (NA). The variance in the accounting base and the materialism of adjustments question both the comparability and reliability of ultimate budgetary balances conducted by EU member states within the EDP requirements (Jesus & Jorge, 2014).

As far as standardisation is concerned, disparities between the accruals-based standards under IPSAS/IFRS/GAAP can be addressed since they share a common genealogy (IFRS from GAAP, IPSAS from IFRS), although certain standards are more particular, such as the ones under IPSAS that are fixated on the public sector. A recent study (Bott & Rüdiger, 2021) from the Hessian Ministry of Finance showed the close link between IPSAS and the German Commercial accounting code, in the context of the EPSAS project. It clearly stated that differences between current national GAAP (HGB) and IPSAS are manageable on the premise that accounting options are used appropriately.

3. Research questions

Statisticians convert source data from different accounting systems into their NA. Thus, the link from commercial and public accounts to NA is not specific to a particular set of accounting standards (whether IFRS, IPSAS, or GAAP). The NA and IPSAS communities regularly discuss how the two standards can be aligned to facilitate the use of PSA information in the compilation of NA (International Public Sector Accounting Standards Board (IPSASB, 2014a).

Vol. 21, No. 4

The discussions for a harmonized integrated accrual-based financial reporting, reconciling public accounts and GFS, have taken place in the EU context, as it is anticipated to broaden the Whole of Government Accounts (WGA) reporting scope. Since September 2014, GA information has been converted to NA in accordance with ESA2010 principles. In general, Eurostat focuses on the compliance of the statistical NA in output terms. There is not a certain requirement on particular inputs to be used, though some are common because these are based on other European statistical legislation. All EU member state governments are obliged to use ESA 2010. Government accounting and in general microeconomic accounting systems of EU member states vary in their source data (administrative, regulatory, surveys etc). From a statistical perspective, despite the variance, the results will follow ESA standards and ensure comparability, and this is checked by the statisticians (Eurostat, 2013). Thus meaning, as far as concerns the ESA2010, there is a recognition on the statistical side that commercial and public accounting approaches should be examined when formulating the statistical approach to an issue. Understanding the relationship between commercial and public accounts and NA is crucial since they are a significant source of data for NA.

This study introduces value relevance models in the public sector to study the impact of the ESA2010 accrual accounting framework on the general government financial statement information of NA in the European Union member states and the UK. The research proposes the analysis of the value of the general government NA for assessing the return on bond prices. The compilation method used for general government NA becomes less statistical and more accounting oriented (Lequiller & Derek, 2007). It investigates whether the accounting framework introduction in statistical financial reporting resulted in financial statement information that better serves its primary objective, which is to provide high quality information on a public entity's performance and financial position both for internal and external stockholders and enable comparability.

Using the ESA2010 statistical indicator, this research employs empirical data in addressing the following research questions to investigate the quality of statistical financial reporting using bond return association models in relevance with the balance sheet, income figures, debt and Euribor figures.

RQ₁: What is the value relevance (thus the quality) of the statistical financial reporting of General Government National Accounts with the ESA2010 accrual accounting framework in EU member states and the UK?

 RQ_2 : Which variables provide stakeholders with high quality governmental statistics reporting that strongly associates and correlates with bonds' return in the EU member states and the UK?

Vol. 21, No. 4

This study contributes empirically, as it introduces value relevance models that measure the quality of the general government NA towards a harmonized methodology with ESA2010 statistical indicator in relevance with decision and policy making and theoretically as it endorses the perception that the accrual accounting basis sets the breeding ground for high quality statistics.

4. Data selection and methodology – Relative value relevance models

Relative value relevance analysis of EU member states and the UK was conducted for the period 1999–2019. In EU member states, the new ESA has been in force since September 2014 (González *et al.*, 2018). The full NA time series in the Eurostat database is on the ESA 2010 basis. Regardless of the year the EU member states join the European Union, countries are legally required to re-transmit back a series of data when a new ESA is introduced, so they work backwards to adjust their data to the new rules. All EU member states published ESA 2010 data (including backwards compatible time series) for the first time in 2014 – this was the legally binding moment for them to do it.

Data for long-term government bond yields were collected from the European Central Bank's statistical data warehouse. Governmental data regarding governments' deficits/surpluses, revenues, expenditures, main aggregates, financial net wealth and government debt, expressed as a percentage of GDP, were derived from the NA indicators (ESA2010 indicator) in the Eurostat database. As the period for which the ESA2010 accrual accounting framework is assessed on its quality of reporting is until 2019, we include the UK in our analysis of EU member states as Brexit occurred in 2020.

The bond return valuation models, apart from variables related to balance sheets and income statements, include Euribor rates. The Euribor (Euro InterBank Offered Rate) is a benchmark rate calculated based on contributions made by a panel of banks, which submit daily an interest rate representing the cost of lending to another large bank, known as the interbank market (Abbassi & Linzert, 2012).

The Euribor rate does not apply to all countries in the study. An example case is the UK, which is not a member of the Euro and therefore the Euribor rate will not apply. However, Barclays UK is among the panel banks and is a primary dealer in Gilts (UK government bonds), U.S. Treasury securities and various European Government bonds. Panel banks ensure that they reflect adequately the diversity of the euro money market and that Euribor represents its underlying market. Direct impact on banks is not the only effect changes in Euribor may produce. This is also an issue for Croatia since it will only join the Euro in 2023. Zaja *et al.* (2018) studied

Vol. 21, No. 4

the determinants of government bond yields in the Republic of Croatia to find that the Euribor as an independent variable has a positive association with interest rates on government bonds and that its growth increases the yield on government bonds.

In this context, the study deems Euribor rates relevant for all sample countries, as these interest rates, within governments are impacted, directly reflect the conditions of an economy and its interbank market. Blommestein *et al.* (2011) underline the increasing reliance of those markets on central banks and government support in both Europe and the United States.

Taking as our reference the equation of Easton and Harris (1991) and Ohlson (1995), we examined the association between return on bonds and financial components in the government sector. For the specification of the first bond return association model and the purposes of this study, the following econometric model is proposed as a basis for estimation and conclusions:

$$BR_{it} = \beta_1 + \beta_2 F E_{it} + \beta_3 (EXP_{it}) + e_{it} \tag{1}$$

where BR_{it} is the annual adjusted return of a 10-year Long-Term Government Bond i the fiscal time period t. We examined bond return values in association with expenditure (EXP) and financial equity (FE) or financial net worth.

Secondly, following Easton *et al.* (2009) and Givoly *et al.* (2017), we proposed the bond return (BR) model as appropriate for relating bond returns to net income figures and the book value buffer as follows:

$$BR_{it} = \delta_{0,t} + \delta_{1,t}NI_{i,t} + \delta_{2,t}\Delta NI_{i,t} + \delta_{3,t}\frac{(FE_{i,t} - D_{i,t})}{D_{i,t}} + \delta_{4,t}Eur_{i,t} + e_{it} (2)$$

where BRr_{it} is the annual return of a 10-year long-term government bond i (in per cent) the fiscal period t. For the independent variables, we use NI (difference in revenue-expenditure or deficit/surplus) and the difference in the net income (Δ NI) within consecutive years. We utilize the cluster of government financial equity (FE) and government debt (D) figures to check the ability of equity to cover government debt (both variables are expressed as a percentage of GDP). The last independent variable is the Euribor rate.

The third regression analysis model evaluates the association of the aforementioned buffer and the Euribor rates with bonds' returns.

$$BR_{it} = \delta_{0,t} + \delta_{1,t} \frac{(FE_{i,t} - D_{i,t})}{D_{i,t}} + \delta_{2,t} Eur_{i,t} + e_{it}$$
(3)

Vol. 21, No. 4

For more details on measuring control variables, we define all the variables in our models in Table 1.

Tab	le 1. Description	on Variables
Variable	Symbol	Source/Measures
Long-term government bond returns	BR	European Central Bank statistical data warehouse
Government financial equity as % of GDP national	FE	National accounts indicators (ESA2010) in Eurostat database
currency Government expenditure as % of GDP in values, national currency	EXP	National accounts indicators (ESA2010) in Eurostat database
Net Income	NI	Government deficit/surplus, revenue, expenditure, and main aggregates, National accounts indicators (ESA2010) in Eurostat database
Difference in NI	ΔΝΙ	Difference is calculated in consecutive years for the period 1999–2019
FE-Debt/Debt	FE- D/ D	Ability of government financial equity to cover government debt. These figures are expressed as % of GDP, National accounts indicators
Euribor rates	Eur	(ESA2010) in Eurostat database. Euribor is short for Euro Interbank Offered Rate. The Euribor rates are based on the average interest rates at which a large panel of European banks borrow funds from one another.

To address the first research question, we investigate the impact of the accounting frameworks as they are shaped throughout the years, between 1999-2019, in EU countries and the UK. We assess the countries' differential response of bonds return to accounting frameworks by comparing their explanatory power (the adjusted R^2) of regressions (1, 2, 3) and the probability value of the variables. The explanatory power measures the relevance (thus quality) of accounting frameworks' information with respect to bond returns, based on the ESA 2010 statistical indicator in the European Union. The bond return association results for each of the subsamples demonstrate which figures, thus independent variables, better interpret the dependent variable for each country. Relative value relevance models enable comparability of the NA statistics quality of the EU member states and the UK.

To address the second research question, the aforementioned models are split into individual analyses. Therefore, using regression and Pearson correlation analysis, we check which coefficient has the best association, thus the ability to best interpret

Vol. 21, No. 4

bonds' return and how bonds' return correlates with individual figures. This analysis enables stakeholders to check financial reports in the European division and see which statistical financial reporting elements, have gradually throughout the period 1999-2019, established strong association and correlation with long-term government bond yields. In our analysis, to determine strong regression and correlation results, we consider 0.7 as the cut-off point that underlines the high quality of financial statistical reporting (Moore *et al.*, 2015). Significance degree analysis is elaborated in Table 2.

 Table 2. Interpretation of association models and Pearson correlation analysis

 Association models analysis

The (R-squared), (also called the coefficient of determination), which is the proportion of variance (%) in the dependent variable that can be explained by the independent variable. Hence, as a rule of thumb for interpreting the strength of a relationship based on its R-squared value (we use the absolute value of the R-squared value to make all values positive)

if R-squared value \leq 0.3 this value is generally considered a None or Very weak effect size,

if R-squared value $0.3 \le r < 0.5$ this value is generally considered a weak or low effect size,

if R-squared value $0.5 \le r < 0.7$ this value is generally considered a Moderate effect size,

if R-squared value $r \ge 0.7$ this value is generally considered strong effect size

Note: Asterisks indicating significance at the 10% (*), 5% (**) and 1% (***) levels

Pearson correlation analysis

Perfect: If the value is near ± 1 , then it said to be a perfect correlation: as one variable increases, the other variable tends to also increase (if positive) or decrease (if negative). High degree: If the coefficient value lies between ± 0.70 and ± 0.90 , then it is said to be a

High degree: If the coefficient value lies between ± 0.70 and ± 0.90 , then it is said to be a strong correlation.

Moderate degree: If the value lies between \pm 0.50 and \pm 0.69, then it is said to be a medium correlation.

Low degree: If the value lies between \pm 0.30 and \pm 0.49, then it is said to be a medium correlation.

Little if any (linear) correlation: When the value is less than ± 0.29

5. Results of statistical reporting quality in the context of ESA2010 accrual accounting framework

5.1 Value relevance (quality) of governmental NA of European Union member states and the UK

Driven by the perception that decision making cannot rely only on cash accounting data, and that public sector accrual accounting practices are gradually considered by both practitioners and researchers (Bergmann *et al.*, 2019), value relevance analysis is performed on the accrual-based accounting frameworks of NA. We investigate whether the quality results justify the efforts for GA and NA alignment and the

Vol. 21, No. 4

standardisation and harmonization process. The requirement for transparency and the significance of GFS in the EU is reflected in ESA 2010 which analyzes its development and the reconciliation of the magnitudes of income, expenditure and net lending/borrowing.

Based on the value relevance results of the first association model, we observe that there is cohesion in the interpretation of financial equity with bonds' return. Out of 27 EU member states and the UK, 23 governments have a significant interpretation of bond return figures with equity coefficient, whilst 17 governments with expenditure one. Governments that fully interpret the model are those of Austria, Croatia, Finland, France, Greece, Hungary, Italy, Latvia, Lithuania, Poland, Portugal, Romania, Slovenia and the UK. In the latter one, estimates and government total public expenditure measurement methods are on an accrual basis. Statistical reporting of Greece and Spain has low explanatory power. The government of Luxemburg and Cyprus are not value relevant. In the majority of EU member states and the UK, stakeholders derive qualitative statistical reporting through financial equity and expenditure figures. Overall, out of the 28 sample countries, 13 governments have a strong association.

From the second value relevance model, net income and Euribor rates have a strong interpretation as 21 governments associate their net income figures with bonds' return and 18 governments associate their Euribor rates with bonds' return. Differences in net income during consecutive years do not provide a strong interpretation of bond returns. This second model differs from the first one, as it includes debt figures and Euribor rates apart from the balance sheet and income figures. The excess of financial equity over government debt is a significant variable only for ten governments. In this second association model, none of the governments, excluding the government of Hungary and Greece.

The third model is simplified, using the Euribor rates and the excess of financial equity over government debt. Again, in this simplified model, Euribor rates have a significant interpretation for 18 out of the 27 EU member states and the UK whilst only 13 governments prove an association of the aforementioned buffer with bonds' return. The governments of Austria, Finland, France, Italy, Lithuania, Slovakia, Spain, and Sweden have a great interpretation of their independent variables with bonds' return. Overall, out of the 28 sample countries, 15 governments have a strong association.

Taking into consideration that we use the relative value relevance approach, the model that best confirms the bonds' return assessment, is the second one recording high quality levels for almost all EU member states and the UK (except for the governments of Greece and Hungary that have a moderate effect). Table 3 illustrates the value relevant results.

Vol. 21, No. 4

n Unio	European Union and the UK?	5											
						. hus					1		
	1 st model					2 nd model	el				3 rd model	el	
B1	+	$\beta_2 F E_{it} + \beta_3 (E X P_{it}) + e_{it}$		BRu =	$BR_{lt} = \delta_{0,t} + \delta_{1,t} N I_{l,t} + \delta_{2,t} \Delta N I_{l,t} + \delta_{3,t}$	$\delta_{2,t}\Delta MI_{i,t} + \delta_{3,t}$	$\frac{\left(FE_{i,t} - D_{i,t}\right)}{D_{i,t}} + \delta_{4,t}Eun_{i,t} + e_{it}$	$_{i,t}Eun_{i,t} + e_{it}$		$BR_{tr} = \delta$	$BR_{it} = \delta_{0,t} + \delta_{1,t} \frac{\left(FE_{i,t} - D_{i,t}\right)}{D_{i,t}}$	$\frac{r}{t}$ + $\delta_{2,t}Ew_{l,t} + e_{lt}$	e _{it}
	FE1	EXP ₂	\mathbb{R}^2	$\delta_{0,t}$	$NI_{i,t}$	$\Delta NI_{i,t}$	$\frac{\left(FE_{i,t}-D_{i,t}\right)}{D_{i,t}}$	Eur _{it}	\mathbb{R}^2	$\delta_{0,t}$	$\frac{\left(FE_{i,t}-D_{i,t}\right)}{D_{i,t}}$	Eur _{it}	\mathbb{R}^2
	0.16***	0.36***	84%	18.98***	-0.36***	0.11	10.72***	0.55***	9/0/6	31.02***	17.41***	0.36**	92%
(06.C)	(10.0)	0.11)		(70°C)	(/0.0)	(0.0)	(56.7)	(0.10)		(6/.C)	(3.42)	(0.12)	
(8.06)	(0.02)	-0.25 (0.14)	38%	-0.90 (5.44)	C0.0- (0.13)	(0.10)	-5.0/ (2.82)	(0.11)	0%06	4.98 (7.39)	1.70 (3.93)	(0.13)	73%
0.20	0.27***	0.08	54%	4.81**	-0.60***	0.22	2.20	0.66	71%	7.02***	3.83*	0.07	43%
-0.35***	0.07***	0.87***	1001	3.46	-0.48***	-0.12	0.57	0.28	1002	(01.2)	(50:2)	(414)	
(9.23)	(0.01)	(0.19)	0/27/	(0.64)	(0.10)	(0.18)	(4.72)	(0.58)	13%0		NK		
	NR			-7.73	-0.47***	0.17*	-6.43	0.80***	72.0%		an		
	1117			(0.21)	(0.09)	(0.08)	(0.10)	(0.19)			1117		
-0.54	0.09***	0.08	0000	2.34	-0.27**	0.04	0.55	0.47	020%	4.84***	2.18**	0.19	1706
(3.76)	(0.01)	(0.08)	0/2/10	(1.53)	(0.11)	(0.10)	(1.10)	(0.29)	04.00	(1.23)	(0.93)	(0.29)	0.6.11
-0.78	-0.10***	0.54	2704	0.54	-0.19**	0.02	-0.63	1.07***	2000	4.98	1.70	0.88***	0404
(6.42)	(0.02)	(0.12)	0/4/2	(0.93)	(0.08)	(0.10)	(0.89)	(0.14)	0/4/0	(0.50)	(0.67)	(0.13)	04.40
-30.32**	0.64***	0.38	170%	-2.01	-1.57***	0.28	1.27^{**}	0.35	940%	1.05	-0.04	1.30*	120%
(10.89)	(0.17)	(0.230	4/ 70	(1.23)	(0.27)	(0.22)	(0.47)	(0.44)	0440	(2.02)	(0.76)	(0.73)	04.04
23.35***	-0.04***	-0.034***	770%	0.85**	-0.30	-0.02	-0.42	1.48***	970%	1.43^{**}	-0.37	0.95***	930%
(3.21)	(0.01)	(0.06)	0/ 7/	(0.37)	(0.17)	(0.09)	(0.43)	(0.30)	0/ /0	(0.23)	(0.47)	(0.10)	02.00
1	0.10^{***}	0.11*	920%	12.07	-0.23	0.10	6.54*	0.64**	010%	20.26***	10.66***	0.32**	9000
(3.29)	(00.0)	(0.06)	04.00	(6.92)	(0.15)	(0.12)	(3.68)	(0.25)	04.12	(4.26)	(2.43)	(0.14)	02.02
-28.84**	-0.01	0.67***	41 0/6	4.15	-0.40***	0.14*	1.97	0.74***	040%	-5.24	-4.08	0.91***	850%
(9.27)	(0.05)	(0.18)	4170	(0.23)	(0.06)	(0.07)	(2.13)	(0.06)	04.04	(0.340	(3.48)	(0.10)	0.20
-12.46	0.06*	0.54**	7406	56.28***	-0.14	0.46	27.90**	-0.55	550%	62.79***	31.18***	0.44	5.10%
(9.74)	(0.03)	(0.21)	0/47	(18.18)	(0.23)	(0.29)	(6.93)	(0.50)	04.00	(13.45)	(7.71)	(0.44)	0440
24.24**	0.07**	0.69***	400%	19.62	-0.26	0.28	8.41	0.19	210%	19.07	7.64	0.45	450%
(10.89)	(0.03)	(0.21)	47.70	(13.52)	(0.24)	(0.26)	(7.23)	(0.42)	0410	(0.11)	(6.15)	(0.36)	04.24
-1.14	0.02	0.16***	400%	0.81	-0.26***	0.23***	-0.21	1.15***	000%	8.54*	3.34	0.19	750%
(1.50)	(0.01)	(0.03)	0/ 24	(2.10)	(0.02)	(0.03)	(1.20)	(0.19)	04.04	(4.80)	(2.84)	(0.42)	01.04
[0.13*	0.85***	64%	43.60***	-0.05	0.08	21.38***	0.28***	82%	44.59***	21.81***	0.27***	820/6
(8 70)	(000)	101 07		100 01	100 00	10.00	101 101	100 07					

Accounting and	Management	Information	Systems
Accounting and		ппот шаноп	Systems

Vol. 21, No. 4

EU MSs	RQ1: What is the in European Unior	RQ1: What is the value relevant in European Union and the UK?	vance (thus th UK?	he qualit	y) of the stat	istical financia	al reporting of	ulty) of the statistical financial reporting of General Government Na	nment Nati	onal Acc	ounts with E	value relevance (thus the quality) of the statistical financial reporting of General Government National Accounts with ESA2010 accrual accounting framework and the UK?	accounting fra	amework
& UK		1 st model	1	8			2 nd model	del				3rd model	el	
Regression Models	$BR_{it} =$	$= \beta_1 + \beta_2 F E_{it} + \beta_3$	$\beta_2 F E_{it} + \beta_3 (EXP_{it}) + e_{it}$		BR _{it} =	$BR_{it} = \delta_{0,t} + \delta_{1,t}NI_{i,t} + \delta_{2,t}\Delta NI_{i,t} + \delta_3$	$-\delta_{2,t}\Delta MI_{t,t} + \delta_{3,t}$	$\frac{\left(FE_{i,t}-D_{i,t}\right)}{D_{i,t}} + \delta_{4,t}Etw_{1,t} + $	$_{4,r}Euv_{1,r} + e_{1r}$		$BR_{tt} = 0$	$BR_{it} = \delta_{0,t} + \delta_{1,t} \frac{\left(FE_{i,t} - D_{i,t}\right)}{D_{i,t}}$	$\frac{1}{2} + \delta_{2,t}Eur_{1,t} + e_{tr}$	+ <i>e</i> _{ir}
Variables/ BR	B1	FEI	EXP ₂	\mathbb{R}^2	$\delta_{0,t}$	NI _{it}	$\Delta NI_{i,t}$	$\frac{\left(FE_{i,t}-D_{i,t}\right)}{D_{i,t}}$	Eur _{it}	\mathbb{R}^{2}	$\delta_{0,t}$	$\frac{\left(FE_{i,t}-D_{i,t}\right)}{D_{i,t}}$	Eur _{it}	R^{2}
Latvia	-28.96*** (5.85)	0.28*** (0.05)	0.91*** (0.15)	72%	2.83* (1.47)	-1.12*** (0.09)	0.19 (0.12)	1.68 (1.15)	0.48 (0.300	929%		NR		
Lithuania	-19.38***	0.10***	0.66***	73%	5.49*	-0.81***	-0.04	2.84	-0.04	77%	-19.38***	0.10***	0.66***	74%
Luxembourg	(22-2)	NR	(71.0)		1.68***	-0.15	-0.11	-0.20	1.21***	82%	1.40***	-0.08	1.00*** (0.21)	80%
Malta	-20.60*** (3.48)	0.02 (0.03)	0.63*** (0.10)	75%	7.41 (5.15)	-0.37*** (0.08)	0.18*	3.04 (2.93)	0.38** (014)	86%	7.22 (7.54)	2.70 (4.28)	0.78*** (0.16)	66%
Netherlands	4.33 (6.94)	0.19*** (0.05)	0.12 (0.16)	44%	1.48 (3.62)	-0.27*** (0.06)	0.02	0.35 (2.18)	0.99*** (0.08)	939%	8.60* (4.30)	4.38 (2.61)	0.83*** (0.10)	84%
Poland	-15.04 (14.25)	0.09* (0.04)	0.52* (0.30)	65%	-3.27 (5.07)	-0.57*** (0.16)	0.09	-3.00 (2.76)	0.99***	80%	8.55* (4.63)	2.76 (2.70)	0.73** (0.26)	63%
Portugal	-11.38* (6.35)	0.04** (0.01)	0.41^{**} (0.14)	36%	62.41*** (8.26)	-0.02 (0.11)	0.07 (0.12)	32.67*** (4.38)	0.14	88%	64.29*** (5.24)	33.63*** (2.93)	0.13 (0.10)	88%
Romania	-9.97** (3.86) -0.44	0.07*** (0.01) 0.13***	0.45*** (0.10) 0.17	84%	5.97*** (0.87) 4.56*	-0.57*** (0.08) -0.41***	0.27** (0.12) 0.11	1.87*** (0.58) 2.12*	-0.03 (0.27) 0.53**	9/006	7.25*** (1.84) 8.77***	1.38 (1.25) 3.67**	0.14 (0.60) 0.48*	43%
Slovania	(4.13) -7.10** (3.14)	(0.01) 0.10*** (0.01	(0.10) 0.24*** (0.06)	78%	(2.17) 8.27* (4.29)	(0.09) -0.39*** (0.10)	(0.08) 0.13 (0.09)	(1.18) 5.68 (3.34)	(0.18) -0.18 (0.80)	0400 179/6	(2.82) 14.46** (5.37)	(1.64) 9.37** (4.30)	(0.26) -1.27 (1.01)	51%
Spain	-1.58 (5.90)	0.05** (0.02)	0.20 (0.14)	25%	5.07 (3.66)	-0.31*** (0.06)	0.31*** (0.08)	2.44 (1.98)	1.04*** (0.15)	84%	10.95* (5.59)	4.61 (3.16)	0.40** (0.18)	44%
Sweden	-2.07 (7.59)	-0.08*** (0.02)	0.11 (0.14)	81%	0.40 (0.34)	-0.28* (0.13)	0.04 (0.11)	-1.86*** (0.62)	0.77*** (0.15)	94%	0.09 (0.33)	-2.52*** (0.54)	0.55*** (0.10)	91%
United Kingdom	17.20*** (2.15)	0.07*** (0.01)	-0.26*** (0.05)	88%	15.10*** (4.5)	0.31*** (0.07)	-0.04 (0.08)	7.02*** (2.29)	0.59*** (0.19)	9/69/6	0.14 (4.52)	-0.14 (2.47)	1.27*** (0.15)	90%
2. X5425	22 22 20	20 10	10 10 10 10 10 10 10 10 10 10 10 10 10 1		100	24. B	1000				20 - 20 - 20	8	8	

Value Relevance of General Government National Accounts with ESA2010 accrual accounting framework. Association of ESA2010 reporting quality with decision making and accounting standardization

Vol. 21, No. 4

5.2 Value relevance and Pearson correlation results of bonds' return association with individual financial figures

To address the second research question, in addition to the aforementioned models, we also analyze the value relevance of bonds' return in governmental NA with the ESA2010 accounting framework individually with financial equity, net income, Euribor, government debt, expenditure, excess of financial equity over government debt (see Table 4). Pearson correlation of individual figures with bond returns is investigated as well (see Table 5). This research perspective intends to highlight the coefficients that correlate with long-term government bond yields, the figures that a stakeholder should examine to interpret the return configuration of bonds and that provide stakeholders with qualitative statistical reporting,

As far as concerns financial equity, there is cohesion in the association with bond yields. Governments that are non-value relevant are those of Cyprus, Germany, Greece, Ireland, Latvia, Luxembourg and Portugal. Based on OECD/International Federation of Accountants (IFAC) (2017) Portuguese and Greek governments were still in the transitioning phase from cash to accrual financial reports in 2016. The rest of the governments extend cash financial reports. Outcomes in Latvia, whose government is based on accrual basis accounting, are irrelevant. During the Eurostat EDP dialogue visit to Latvia (EC, 2019) Eurostat highlighted the need for the compilation of annual financial accounts based on ESA rules, prioritizing the GA over other data sources. Results that provide qualitative statistical reports concern the governments of Austria, the Czech Republic, France, Slovakia, Sweden and the UK. Their Pearson correlation is also of a high degree. Financial equity in the Netherlands has a qualitative and significant association with bonds' return but not from an income perspective. In the Netherlands, the statistical office prepares the balance sheet that relies on entities' individual financial reports. These also encompass national public agencies, whose accounts base on an accrual basis.

Income association figures demonstrate that the governments of Belgium, Bulgaria, Denmark, Finland, France, Ireland, the Netherlands and Spain are not value relevant. The same applies to Bulgaria, Finland, the Netherlands and Spain from an expenditure approach. Even though the governments of Belgium, Denmark, France and Ireland are not value relevant from the net income approach, these are relevant from the expenditure approach. In Ireland, the budget includes fiscal estimates for general government revenue and expenditure prepared in accordance with the European Statistical Standard ESA 2010. Conversely, Cyprus, Estonia, Portugal, Slovakia and Slovenia are not value relevant from an expenditure perspective but value relevant from a net income approach. Pearson correlation results for these two coefficients range from low to moderate scale. The difference in net income is not relevant for EU member states and the UK (therefore not presented in Table 4) and also has no linear correlation in the EU division (see Table 5).

Vol. 21, No. 4

Value Relevance of General Government National Accounts with ESA2010 accrual accounting framework. Association of ESA2010 reporting quality with decision making and accounting standardization

Regarding governments' debt coefficient, 19 governments interpret bonds' return. Excess financial equity over government debt coefficient is confirmed for 22 governments out of 28. France, Luxembourg and the UK not only are relevant but have the highest explanatory power in both variables. In France, accrual-based data is used to establish the NA (statistics), albeit after several restatements. Corresponding to these two variables, the average correlation analysis for EU member states and UK is of little and low degree.

The governments of Austria, Czech Republic, Lithuania, Malta, Romania, Sweden and the UK have qualitative statistical reporting and association with bonds' return for all coefficients. Respectively, a high degree of correlation is evidenced in the governments of Austria, the Czech Republic, Romania and the UK which follow an accrual basis in governmental accounting for over a decade. Value relevance results hereby endorse the efforts of Austria and the UK to harmonize the accounting basis and coverage of fiscal reports (budget, financial statements, and statistics). This harmonization allows greater use of the accounting data for financial analysis and greater transparency of the state of public finances (OECD/IFAC, 2017) and this is hereby confirmed. In the case of Romania, the consolidation of subsidies, transfers and interests, non-financial and financial accounts are made by the Ministry of Finance, the National Institute of Statistics and the National Bank of Romania, following the ESA 2010 requirements. The Swedish government which proves to generate qualitative statistical reporting, measures the net lending and budget balance, based on statistical standards. As OECD/IFAC (2017) highlights, these are key fiscal elements that draw most the political and public consideration.

Vol. 21, No. 4

EU MSs	$\mathbf{Final}_{lr} = \delta_{0_i}$	Financial Equity $BR_{lt} = \delta_{0,t} + \delta_{1,t}FE_{l,t} + \epsilon$	lity r + e _{ir}	$\mathbf{P}_{it} = \delta$	Net Income $BR_{it} = \hat{a}_{0,t} + \hat{a}_{1,t}MI_{i,t} + e_{it}$	+ e _{ir}	$BR_{tr} = \delta_{0,t}$	Euribor $BR_{it} = \delta_{0,t} + \delta_{1,t}Euv_{1,t} + e_{it}$	+ e _{lt}	$\mathbf{Gov}_{ic} = \delta_{0,c}$	Government debt $BR_{it} = \delta_{0,t} + \delta_{1,t}G.Debt_{i,t} + e_{it}$	ebt $t_{i,t} + e_{it}$	$\mathbf{E}\mathbf{X}_{it} = \delta_{0,t}$	Expenditure $BR_{it} = \delta_{0,t} + \delta_{1,t} EXP_{i,t} + e_{it}$	⊢ e _{it}	Excess c $BR_{tt} = \delta_{0,t}$	Excess of FE over G. Debt $BR_{tr} = \delta_{0,t} + \delta_{tx} \frac{(FE_{tx} - D_{tx})}{D_{tx}} + \epsilon_{tx}$	Debt + e _{it}
Variables	$\delta_{0,t}$	Fe	\mathbb{R}^{2}	$\delta_{0,t}$	IN	\mathbb{R}^2	$\delta_{0,t}$	Eur	R ²	$\delta_{0,t}$	Debt	\mathbb{R}^2	$\delta_{0,t}$	Exp	\mathbb{R}^2	$\delta_{0,t}$	(FE-Debt)/ G. Debt	\mathbb{R}^{2}
Austria	11.22*** (1.12)	0.17*** (0.02)	74%	1.17*** (0.57)	-0.71*** 0.20)	35%	1.40*** (0.24)	0.95*** (0.10)	79%	15.00*** (3.28)	-0.16*** 0.04	35%	-30.40*** (8.73)	0.65*** (0.16)	39%	45.18*** (3.63)	25.67*** (2.21)	87%
Belgium	-4.09 (2.44)	-0.08***	40%	-	NR		1.63*** (0.27)	0.90*** (0.12)	73%	-5.89 (3.70)	0.08** (0.03)	21%	18.62** (8.58)	-0.28* (0.16)	11%	-19.56 (12.49)	-12.16* (6.54)	13%
Bulgaria	3.20*** (0.38)	0.27***	53%		NR	2	2.92*** (0.51)	0.79** (0.28)	32%	4	NR	3.	NR	æ		7.35*** (1.06)	4.12*** (1.21)	43%
Croatia	6.06*** (0.05)	0.05*	26%	2.72*** (0.45)	18	67%	Z	NR	MR	7.54*** (1.77)	-0.50* (0.02)	20%	-27.05* (13.97)	0.66** (0.29)	28%		NR	
Cyprus	NR	R	NR	4.04*** (0.42)	-0.27** (0.10)	30%	3.96*** (0.48)	0.46* (0.23)	18%	4	NR	R	NR	æ	NR		NR	
Czech Republic	3.08*** (0.17)	0.09***	78%	1.99***	-0.49*** (0.10)	55%	2.01*** (0.29)	0.83*** (0.13)	69%	8.69*** (1.59)	-0.16*** (0.04)	42%	-10.97 (6.85)	0.33* (0.16)	20%	5.62*** (0.38)	2.74*** (0.37)	76%
Denmark	2.11*** (0.39)	-0.11***	56%	-	NR		1.28*** (0.23)	0.99***	83%	-2.93** (1.32)	0.14*** (0.02)	50%	-11.71 (8.38)	0.28* (0.15)	12%	-1.82 (1.96)	-4.06** (1.70)	24%
Estonia	-15.13** (5.52)	0.62*** (0.17)	40%	4.23*** (0.84)	-0.98* (0.51)	16%	0.94 (0.860	1.26*** (0.35)	43%	9.47*** (2.75)	-0.81** (0.39)	19%	NR	8		-1.96 (2.01)	1.44*** (0.47)	34%
Finland	6.41*** (1.26)	-0.06** (0.02)	28%	1	NR		1.34*** (0.22)	0.97*** (0.10)	84%	9.65*** (2.13)	-0.12*** (0.04)	27%	NR	~			NR	
France	7.85*** (0.25)	(00.0)	95%		NR		1.49*** (0.23)	0.90*** (0.10)	79%	12.02*** (0.75)	-0.11*** (0.00)	85%	36.24*** (8.97)	-0.59*** (0.16)	37%	29.26*** (2.42)	15.72*** (1.45)	85%
Germany	NR	К		2.14** (0.30)	-0.63 (0.10)	62%	1.02*** (0.22)	1.00*** (0.99)	83%	4	NR		-24.47*** (5.27)	0.59*** (0.11)	55%	-25.86** (10.69)	-18.17** (6.74)	26%
Greece	NR	R		5.35*** (1.71)	-0.39* (0.22)	12%	N	NR	s2	Ā	NR		-12.02 (10.16)	0.40* (0.20)	14%	65.83*** (15.57)	32.79*** (8.80)	38%
Hungary	10.03*** (2.11)	0.07* (0.03)	17%	3.91*** (0.88)	-0.49** (0.17)	32%	4.75*** (0.50)	0.88*** (0.24)	43%	4	NR		-28.37** (12.04)	0.70** (0.24)	33%	30.21*** (7.16)	13.55*** (4.04)	40%
Ireland	NR	ы		1	NR		2.68***	0.65**	23%	A	NR		-1.93	0.17***	49%	11.23***	4.78***	35%

Vol. 21, No. 4

	Financial Equity Net Income Eurlbor Government debt Expenditure Excess of F	Financial Equity	A	N	Net Income			Euribor		Gove	Government debt	bt	Ext	Expenditure		Excess of	Excess of FE over G. Debt	Debt
EU MSs	$BR_{it} = \delta_{0}$	$BR_{it} = \delta_{0,t} + \delta_{1,t}FE_{i,t} + \delta_{1,t}F$	+ e _{tr}	$BR_{tr} = \delta$	$BR_{it} = \delta_{0,t} + \delta_{i,t}NI_{i,t} + e_{it}$	+ e _{ir}	$BR_{it} = \delta_{0,t}$	$BR_{it} = \delta_{0,t} + \delta_{1,t}Euv_{i,t} + s_{it}$	$+ e_{ir}$	$BR_{it} = \delta_{0,t}$	$BR_{it} = \delta_{0,t} + \delta_{1,t}G.Debt_{i,t} + e_{it}$	$a_{tr} + a_{tr}$	$BR_{tr} = \delta_{0,r}$	$BR_{it} = \delta_{0,t} + \delta_{1,t} EXP_{i,t} + e_{it}$	+ e _{ir}	$BR_{it} = \delta_{0,t} +$	$BR_{it} = \delta_{0,t} + \delta_{i,t} \frac{\left(FE_{i,t} - D_{i,t}\right)}{D_{i,t}} + e_{it}$	+ e _{ir}
Variables	$\delta_{0,t}$	Fe	\mathbb{R}^2	$\delta_{0,t}$	IN	\mathbb{R}^2	$\delta_{0,t}$	Eur	\mathbb{R}^2	$\delta_{0,t}$	Debt	\mathbb{R}^2	$\delta_{0,t}$	Exp	\mathbb{R}^2	$\delta_{0,t}$	(FE-Debt)/ G. Debt	\mathbb{R}^2
Italy	14.73*** (3.04)	0.09*** (0.02)	33%	0.62 (0.84)	-1.22*** (0.24)	53%	2.93*** (0.29)	0.55*** (0.13)	47%	12.73*** (4.29)	-0.07*	13%	NR	NR	NR	82.89*** (13.07)	41.63*** (6.95)	%09
Latvia		NR		2.14*** (0.53)	-1.03*** (0.15)	72%	3.27*** (0.91)	0.84* (0.42)	19%	N	NR	NR	NR		NR		NR	NR
Lithuania	4.93*** (0.71)	0.10** (0.04)	%57	2.44*** (0.62)	-0.81*** (0.16)	57%	2.79*** (0.84)	0.96** (0.40)	24%	8.33*** (2.17)	-0.14* (0.06)	19%	-20.63*** (5.86)	0.68*** (0.16)	51%	(95.1) (1.56)	2.70* (1.34)	19%
Luxembourg	NR	NR	NR	2.42*** (0.58)	0.43* (0.22)	14%	1.14*** (0.25)	0.95*** (0.11)	78%	6.58*** (0.66)	-0.24*** (0.04)	58%	22.18*** (7.82)	-0.45** (0.18)	20%	0.76 (0.51)	0.64*** (0.13)	52%
Malta	-1.55 (2.45)	-0.11* (0.05)	21%	2.40*** (0.28)	-0.51*** (0.07)	72%	2.30*** (0.31)	0.88*** (0.14)	%99	-6.83*** (1.73)	0.16*** (0.02)	67%	-19.98*** (3.33)	0.58*** (0.08)	75%	23.88** (10.01)	11.66* (5.76)	19%
Netherlands	9.35*** (1.67)	0.18*** (0.04)	42%	NR	NR	NR	1.26*** (0.23)	0.96*** (0.10)	81%	NR	NR	NR	NR	NR	NR	24.82*** (7.66)	13.56** (4.75)	29%
Poland	9.97*** (1.01)	0.16*** (0.03)	29%	2.87*** (0.83)	-0.59*** (0.19)	35%	3.64*** (0.38)	0.99*** (0.18)	61%	18.43*** (4.39)	-0.27*** (0.08)	37%	-37.63*** (8.83)	0.98*** (0.20)	58%	18.21*** (3.42)	8.13*** (2.12)	46%
Portugal	N	NR		3.30*** (1.04)	-0.38* (0.19)	14%	N	NR		Z	NR		NR	~		64.12*** (5.37)	33.39*** (3.01)	85%
Romania	6.66*** (0.35)	0.09*** (0.02)	%09	4.16*** (0.67)	-0.52*** (0.16)	44%	5.16*** (0.44)	0.78*** (0.25)	40%	9.48*** (1.05)	-0.12*** (0.03)	48%	-15.59** (6.70)	0.59*** (0.18)	45%	7.66*** (0.65)	1.65*** (0.52)	43%
Slovakia	6.59*** (0.52)	0.12*** (0.01)	72%	1.24 (0.72)	-0.63*** 0.16	46%	3.78*** 0.39)	0.73*** (0.23)	44%	8.33*** (2.52)	-0.10* (0.05)	18%	NR			12.92*** (1.75)	5.95*** (1.14)	64%
Slovenia	4.42*** (0.36)	0.10*** (0.02)	%65	3.09*** (0.65*)	-0.24* (0.13)	17%	2.57*** (0.54)	0.94*** (0.28)	39%	6.82*** (1.04)	-0.06*** (0.01)	39%	NR			6.88*** (1.43	2.56* (1.34)	20%
Spain	6.84*** (1.03)	0.05** (0.01)	25%	NR	NR	NR	2.61*** (0.37)	0.62*** (0.16)	41%	7.52*** (1.38)	-0.04** (0.01)	21%	NR			4	NR	
Sweden	3.78*** (0.19)	-0.10*** (0.10)	80%	3.67*** (0.44)	-0.55** (0.21)	23%	1.33*** (0.23)	0.94*** (0.10)	80%	-5.52*** (1.24)	0.19*** (0.02)	70%	-27.20*** (3.33)	0.58*** (0.06)	79%	1.31 (1.08)	-2.59** (1.09)	19%
United Kingdom	7.51*** (0.57)	0.09*** (0.01)	%9L	5.34*** (0.64)	0.54*** (0.13)	43%	0.40 (0.24)	1.26*** (0.10)	%06	9.15*** (0.63)	-0.10*** (0.01)	82%	23.36*** (3.53)	-0.49*** (0.08)	%09	33.93*** (4.26)	17.45*** (2.42)	71%
Note: ΔM is not value relevant and therefore omitted	ot value rele	vant and th	erefore	omitted.														

Value Relevance of General Government National Accounts with ESA2010 accrual accounting framework. Association of ESA2010 reporting quality with decision making and accounting standardization

Vol. 21, No. 4

EU MSs	Financial Equity	Net Income	Euribor	Gov't Debt	Expend.	Diff. in Net Income	(FE- DEBT)/ DEBT
Austria	0.86	-0.59	0.89	-0.59	0.63	-0.27	0.94
Belgium	-0.55	0.04	0.89	0.44	-0.41	-0.26	-0.41
Bulgaria	0.73	-0.39	0.57	-0.05	0.10	-0.47	0.66
Croatia	0.51	-0.82	0.43	-0.45	0.53	-0.15	0.42
Cyprus	0.23	-0.55	0.43	-0.35	0.09	-0.25	0.20
Czech Republic	0.89	-0.75	0.84	-0.60	0.45	-0.22	0.87
Denmark	-0.75	-0.15	0.91	0.74	0.35	-0.34	-0.49
Estonia	0.63	-0.40	0.67	-0.44	0.20	-0.10	0.59
Finland	-0.53	0.18	0.91	-0.52	-0.05	-0.31	0.10
France	0.98	0.03	0.89	-0.92	-0.61	-0.16	0.92
Germany	0.10	-0.79	0.91	-0.33	0.74	-0.09	-0.51
Greece	0.27	-0.35	-0.27	0.10	0.38	-0.01	0.61
Hungary	0.42	-0.57	0.66	-0.22	0.57	0.16	0.63
Ireland	0.22	-0.29	0.48	0.15	0.63	-0.45	0.59
Italy	0.57	-0.73	0.69	-0.37	0.30	-0.32	0.78
Latvia	0.37	-0.85	0.47	-0.17	0.42	-0.47	0.32
Lithuania	0.50	-0.76	0.49	-0.44	0.72	0.06	0.44
Luxembourg	0.25	0.37	0.88	-0.76	-0.45	-0.10	0.73
Malta	-0.46	-0.85	0.81	0.82	0.86	-0.23	0.44
Netherlands	0.65	-0.33	0.90	0.09	0.32	-0.24	0.54
Poland	0.77	-0.60	0.79	-0.60	0.76	-0.10	0.68
Portugal	0.29	-0.38	0.20	-0.24	0.09	-0.22	0.93
Romania	0.77	-0.67	0.63	-0.70	0.67	0.13	0.65
Slovak Republic	0.85	-0.68	0.66	-0.42	-0.07	0.05	0.80
Slovenia	0.77	-0.41	0.63	-0.63	0.41	0.24	0.45
Spain	0.49	-0.21	0.64	-0.46	0.12	0.29	-0.33
Sweden	-0.90	-0.48	0.90	0.84	0.89	-0.29	-0.44
United Kingdom	0.87	0.66	0.95	-0.91	-0.78	0.03	0.84
Average of EU	0.35	-0.40	0.67	-0.25	0.28	-0.15	0.43

 Table 5. Pearson Correlation analysis

 RQ2: Which variables provide stakeholders with financial statistical reporting that best correlate with bonds' return in EU MSs and the UK?

Vol. 21, No. 4

6. Discussion and implications of the reporting quality results in association with decision making and accounting standardisation

6.1 Association of ESA2010 reporting quality with decision and policymaking

Statistics undoubtedly play a significant role in the international economy and are an important resource for policymakers, business leaders and the general public. They support evidence-based decision making, allow comparisons over time between policies, countries and economies, and help increase transparency and accountability. Statistical indicators such as ESA2010 must be commonly accepted by EU member states and based on a methodology of common acceptance. European statistics, as produced and disseminated by Eurostat and the EU national statistical institutes, meet the highest quality requirements. The fact that the ESA2010 accounting framework relies on an accrual accounting basis is itself a precondition for high quality information for fruitful decision making. Previous studies have globally perceived the usefulness of accrual accounting based financial information for accountability and for supporting decision making in public sector organizations (Andriani et al., 2010; Ismail, 2022; Kober et al., 2010). However, relative value relevance is examined in the European accrual accounting framework to affirm its quality and usefulness, considering that other authors (Brusca & Montesinos, 2013; Caruana & Farrugia, 2018) find higher usefulness of cash data.

Further to this approach, the basis of high-quality statistics is not the absolute precondition for making relevant indicator policy and statistical reports. Formal statisticians, researchers and experts need to be aware of the impact of the growing role of statistical reporting in policymaking and adapt the statistical indicators and accounting frameworks that justify the methodology from which they emerge. They need to work with decision makers to understand their needs and expectations, but also to guide the correct interpretation and use of these indicators, as well as their limitations. The growing role of statistics in policymaking and decision making requires quality assurance, even if the primary data source arises from different accounting methods (cash or accrual) but also from different accrual-based standards under IFRS/IPSAS/GAAP. This research study demonstrates econometric models that confirm the high quality of the NA of General Government financial statements of EU member states and the UK through their indicators, which is in line with stakeholders' expectations for a high standard of excellence from NA. However, this study shows which data are most important for better decision making by decisionpolicy makers and all pertinent stakeholders since indicators need to provide information that fulfils individual needs, be precisely assessed, fast, comprehensive, and simple.

Vol. 21, No. 4

6.2 Association of ESA2010 reporting quality with accounting standardisation

In the last years, there has been a continuous two-way effort by the statistical and accounting community to converge macro and micro-economic government accounting information through harmonized, integrated reporting on an accrual basis (EC, 2013a; IPSASB, 2014a). Bracci *et al.* (2015) underlined that in most European countries, cash accounting persisted at the centre of the budget process, which prevailed over financial reporting in its significance (Heiling *et al.*, 2013).

The variance between financial and statistical reports is traced in terms of measurement, timeline, recording and conceptual framework (Chan & Heiling, 2012). At the European level, the alignment effort is reflected in the harmonization of IPSAS bv those governing the European system (ESA2010). PricewaterhouseCoopers's (PwC's) (2014) survey on behalf of Eurostat, confirmed the suitability of IPSAS as a reference to developing EPSAS in EU member states (EC, 2013b, 2014) which in turn can contribute to improving the reliability of ESA. In the EU, IPSAS as the reference point of EPSAS development was initiated to achieve a homogeneous EU-landscape of PSA governance and reporting (Lorson et al., 2019) aimed at harmonization (Pontoppidan & Brusca, 2016). EU authorities are creating policies and projects like EPSAS, based on the SNA, in an effort to close the information gap between GA and NA (Dasí et al., 2013). The integration of national accounts, budgetary and financial reporting standardizes conciliation on their information, facilitates the internal use of national accounts and constitutes EPSAS a comparative source among EU countries (Manes-Rossi et al., 2016).

In the EPSAS workshop that took place in Brussels (Directorate-General for Internal Policies - Policy Department D: Budgetary Affairs, 2015), it was highlighted the interconnection of EPSAS with the ESA as a facilitator of translating public sector accounts into financial statistics which would enable Eurostat to check fiscal data more easily. The same workshop considered that GFS would improve its quality on the premise that the interconnectedness of PSA and ESA strengthens, setting prerequisites that EPSAS base on ESA and suggesting changes to ESA where needed. Verrinder (2021) endorses the EC proposal (EC, 2014) of the Framework Regulation which includes fundamental EPSAS principles such as accrual-based accounting, double entry book-keeping and therefore endorses accounting standards consistent with ESA principles, having IPSAS as the first reference base. Considering EPSAS, the importance of the alignment of GA with the NA, namely ESA2010, became prevalent, considering that the figures from the former are input for the latter, based on which, the EU fiscal discipline and macroeconomic convergence criteria are assessed (Jorge et al., 2019). Essentially, financial reporting and statistical reporting have separate yet complementary functions in terms of the interconnection of EPSAS and ESA2010. Both are needed, EPSAS cannot be

Vol. 21, No. 4

expected to fully align with ESA but by utilizing IPSAS convergence efforts with NA, some differences may be eliminated.

Value relevance results endorse the quality and therefore usefulness of statistical reporting information to decision and policymakers. The NA proves to be a rich statistical source, with legislated rules in Europe (ESA 2010) to ensure comparability. Given the importance of statistical reporting to policymakers, debt managers, ratings agencies and investors in determining the progress being achieved in an economy, the contribution of the ESA accounting framework proves to be of high quality resulting in statistical reporting of great value relevance, reliability, and perceiving the interconnection efforts of accounting standardisation with ESA2010 to improve their mutual quality.

7. Conclusions

Financial reporting and statistical accounting are studied by Heald and Hodges (2018) as independent GA approaches that have been integrated under the influence of international agencies. The GFS is capable of providing a unified structure of the whole economy, its components, and its relationships with other total economies owing to the international standards ESA2010 and SNA2008 (Lorson *et al.*, 2019).

High relative value relevance results in the majority of sample governments, clearly demonstrate the quality of statistical reporting, perceiving the evident European efforts for the "unconcealed ambition to build a uniform accrual-based budgeting and accounting system for all EU member states" to achieve unification in the context of the EPSAS project (Mussari, 2014). This study does not overlook the discrepancies between GA and NA. It does not connive that envisaging compilation of GFS as a result of expedient IPSAS reconciliation entails awareness of issues of entity concept, consolidation, recognition, and measurement valuation. Neither it implies that EPSAS can be expected to fully align with ESA.

This study investigates the quality of GFS reporting with the ESA2010 accrual accounting framework. It presents the theoretical and practical implications of national account information on evaluating bond returns, constructed on global standards of accrual accounting premise. It considers an incentive the contribution of accrual accounting frameworks to the quality of statistical reporting for the interconnectedness of EPSAS with ESA2010 when formulating standards (IPSASB, 2014b). This study endorses harmonization efforts of EPSAS with the ESA framework as the alignment of micro-macro will lead to ESA changes where necessary fostering both financial and statistical reporting quality. In line with the 2015 EPSAS workshop (Directorate-General for Internal Policies – Policy Department D: Budgetary Affairs, 2015), it endorses the efforts for harmonizing the

Vol. 21, No. 4

accounting basis and coverage of fiscal reports (budget, financial statements, and statistics) and proves that indeed this harmonization has allowed greater use of the accounting data for fiscal analysis and greater transparency of the state of public finances and also proves that in such cases (e.g., governments of Austria and the UK) provides fiscal reporting of high quality (OECD/IFAC, 2017).

This study underlines ESA2010 as a significant event in NA, further validating their critical economic function, and emphasising the calibre of their reporting. It is in line with Caruana *et al.* (2019), that despite the different objectives of NA, budgetary and financial reporting, aggregated systems should enable comparability. These should be integrated into a comprehensive financial and management information system (Barton, 2011) that produces adequate reports on governments' performance.

Evidence of value relevance analysis proves that cash-based governments yield qualitative results when converted to accrual, confirming the statistical perspective, that despite the variance in input terms, results will follow ESA standards and ensure comparability facilitating policy and decision making. As budgetary and accrual accounting systems have been integrated to a great extent and contain budgetary and ESA 2010 codes, an account of budget implementation needs to be drafted, complementary to the financial statement reporting (Christiaens & Vanhee, 2007).

With this study, there are opportunities for accounting academics and practitioners to research the quality of public sector governmental accounting comparative analysis with corresponding statistics, using promoted value relevance tools. Bergmann (2021), underlines that "if financial statements do matter economically, their quality should also matter" and approaches this issue from an audit surveillance perspective ensuring their quality. Indeed, improvements in government and national accounts reporting alignment efforts should always strive to improve their reporting quality proactively to facilitate both reporting and audit purposes.

Acknowledgements

Authors wish to acknowledge their appreciation to the peer reviewers for the overall evaluation, the insightful input and valuable suggestions.

References

- Abbassi, P. & Linzert, T. (2012) "The effectiveness of monetary policy in steering money market rates during the financial crisis", *Journal of Macroeconomics*, vol. 34, no. 4: 945-954
- Andriani, Y., Kober, R. & Ng, J. (2010) "Decision usefulness of cash and accrual information: public sector managers' perceptions", *Australian Accounting Review*, vol. 20, no. 2: 144-153

Vol. 21, No. 4

Value Relevance of General Government National Accounts with ESA2010 accrual accounting framework. Association of ESA2010 reporting quality with decision making and accounting standardization

- Ball, R. & Brown, P. (1968) "An empirical evaluation of accounting income numbers", *Journal of Accounting Research*, vol. 6, no. 2: 159-178
- Ball, R. & Shivakumar, L. (2008) "Earnings quality at initial public offerings", *Journal of Accounting and Economics*, vol. 45, no. 2-3: 324-349
- Barth, M.E., Beaver, W.H. & Landsman, W.R. (2001) "The relevance of the value relevance literature for financial accounting standard setting: another view", *Journal of Accounting and Economics*, vol. 31, no. 1-3: 77-104
- Barth, M.E., Landsman, W.R., Lang, M. & Williams, C. (2012) "Are IFRS-based and US GAAP-based accounting amounts comparable?", *Journal of Accounting and Economics*, vol. 54, no. 1: 68-93
- Barton, A. (2011) "Why governments should use the government finance statistics accounting system", *Abacus*, vol. 47, no. 4: 411-445
- Bergmann, A. (2021) "Editorial: quality of financial information presented by public sector entities", *Public Money & Management*, vol. 41, no. 8: 581
- Bergmann, A., Fuchs, S. & Christoph, S. (2019) "A theoretical basis for public sector accrual accounting research: current state and perspectives", *Public Money and Management*, vol. 39, no. 8: 560-570
- Blommestein, H.J., Keskinler, A. & Lucas, C. (2011) "Outlook for the securitisation market", OECD Journal: Financial Market Trends, vol. 2011, no. 1, available on-line at https://www.oecd.org/daf/fin/financial-markets/48620405.pdf (Retrieved November 15, 2022)
- Bott, H. & Rüdiger, G. (2021) "IPSAS financial statements 2019 of the Federal State of Hessen: eleventh meeting of the EPSAS Working Group (28 / 29 April 2021)", Hessian Ministry of Finance, available on-line at https://circabc.europa.eu/sd/a/d5e7f3e7-c74e-45be-8724ba8f9de11400/Agenda%20item%204%20-%20DE Pr%25c3%25a4sentation%2028.04.2021%20III III%209.pptx final

ENG.pdf (Retrieved November 15, 2022)

- Bracci, E., Humphrey, C., Moll, J. & Steccolini, I. (2015) "Public sector accounting, accountability and austerity: more than balancing the books?", Accounting, Auditing & Accountability Journal, vol. 28, no. 6: 878-908
- Brito, J.R. & Jorge, S. (2021) "The institutionalization of a new accrual-based public sector accounting system: the case of Cape Verde", *International Journal of Public Administration*, vol. 44, no. 5: 372-389
- Brusca, I. & Montesinos, V. (2013) "From rhetoric to practice: the case of Spanish local government reforms", *Financial Accountability and Management*, vol. 29, no. 4: 267-287
- Caruana, J., Dabbicco, G., Jorge, S. & Jesus, M.A. (2019) "The development of EPSAS: contributions from the literature", *Accounting in Europe*, vol. 16, no. 2: 146-176
- Caruana, J. & Farrugia, B. (2018) "The use and non-use of the government financial report by Maltese members of parliament", *Accounting, Auditing & Accountability Journal*, vol. 31, no. 4: 1124-1144

Vol. 21, No. 4

- Chan, J. & Heiling, H. (2012) "From servant or master? On the evolving relationship between accounting and budgeting in the public sector", 2012 Yearbook of the Swiss Society of Administrative Sciences, 23-38
- Christiaens, J. & Vanhee, C. (2007) Handbook Accountancy in the Public and Non-Profit Sector, Audit Accountancy Tax, Brugge: Die Keure
- Christiaens, J., Vanhee, C., Manes-Rossi, F., Aversano, N. & van Cauwenberge, P. (2015) "The effect of IPSAS on reforming governmental financial reporting: an international comparison", *International Review of Administrative Sciences*, vol. 81, no. 1: 158-177
- Clarkson, P., Hanna, J.D., Richardson, G.D. & Thompson, R. (2011) "The impact of IFRS adoption on the value relevance of book value and earnings", *Journal of Contemporary Accounting & Economics*, vol. 7, no. 1: 1-17
- Dasí, R.M., Montesinos, V. & Murgui, S. (2013) "Comparative analysis of governmental accounting diversity in the European Union", *Journal of Comparative Policy Analysis: Research and Practice*, vol. 15, no. 3: 255-273
- Directorate-General for Internal Policies Policy Department D: Budgetary Affairs (2015) "EPSAS European Public Sector Accounting Standards How to achieve more reliable and transparent accounting systems: how to design EPSAS for a better implementation of the EU budget in the Member States", available on-line at https://www.europarl.europa.eu/RegData/etudes/STUD/2015/552299/IPOL_STU(2015)552299_EN.pdf (Retrieved November 15, 2022)
- Easton, P.D. & Harris, T.S. (1991) "Earnings as an explanatory variable for returns", *Journal of Accounting Research*, vol. 29, no. 1: 19-36
- Easton, P.D., Monahan, S.J. & Vasvari, F.P. (2009) "Initial evidence on the role of accounting earnings in the bond market", *Journal of Accounting Research*, vol. 47, no. 3: 721-766
- European Commission (EC) (2011), "Council directive 2011/85/EU, on requirements for budgetary frameworks of the member States, L306/41", *Official Journal of the EU*.
- European Commission (EC) (2012) "Public consultation: assessment of the suitability of the IPSAS for the Member States", available on-line at https://ec.europa.eu/eurostat/documents/4187653/5797841/D4_2012-EN.PDF.pdf/4270cc6e-8c69-455e-97f9-2255aefacd41?t=1414778283000 (Retrieved November 15, 2022)
- European Commission (EC) (2013a) "Commission staff working document accompanying the document Report from the Commission to the Council and the European Parliament – Towards implementing harmonised public sector accounting standards in Member States: the suitability of IPSAS for the Member States", available on-line at https://eur-lex.europa.eu/legalcontent/EN/TXT/PDF/?uri=CELEX:52013SC0057&from=EN (Retrieved November 15, 2022)

- European Commission (EC) (2013b) "Report from the Commission to the Council and the European Parliament – Towards implementing harmonised public sector accounting standards in Member States: the suitability of IPSAS for the Member States", available on-line at https://eur-lex.europa.eu/legalcontent/EN/TXT/PDF/?uri=CELEX:52013DC0114&from=EN (Retrieved November 15, 2022)
- European Commission (EC) (2014) "Task Force EPSAS Governance. Public consultation on future EPSAS governance principles and structures: draft report", available on-line at https://circabc.europa.eu/sd/a/15f43dec-5284-46fd-98a2-f7cc10f0fd78/Agenda%20item%202%20-%20Draft%20Notes%20TF%20EPSAS%20Gov%2027%20March.pdf (Retrieved November 15, 2022)
- European Commission (EC) (2019) "Final findings: Eurostat EDP dialogue visit to Latvia, 2-4 December 2019", available on-line at https://ec.europa.eu/eurostat/ documents/1015035/9983802/Final-findings-EDP-dialogue-visit-LV-2-4-Dec-2019.pdf/4dcea88b-24c1-692b-7a03-0a1614a44c32 (Retrieved November 15, 2022)
- Eurostat (2013) "European system of accounts: ESA 2010", available on-line at https://ec.europa.eu/eurostat/documents/3859598/5925693/KS-02-13-269-EN.PDF/44cd9d01-bc64-40e5-bd40-d17df0c69334 (Retrieved November 15, 2022)
- Eurostat (2014) "Manual on the changes between ESA 95 and ESA 2010", available on-line at https://ec.europa.eu/eurostat/documents/3859598/5936825/KS-GK-14-002-EN-N.pdf/059d43eb-2047-40c3-9cdd-d847bfbce33b (Retrieved November 15, 2022)
- Francis, J. & Schipper, K. (1999) "Have financial statements lost their relevance?", *Journal of Accounting Research*, vol. 37, no. 2: 319-352
- Ghani, E.K., Jusoh, W.N.H.W., Hassan, R. & Muhammad, K. (2019) "Determinants of auditors' readiness for accrual accounting adoption", *International Journal* of Recent Technology and Engineering, vol. 8, no. 3: 8664-8673
- Givoly, D., Hayn, C. & Katz, S. (2017) "The changing relevance of accounting information to debt holders over time", *Review of Accounting Studies*, vol. 22: 64-108. doi:10.1007/s11142-016-9374-y
- Gomes, P.S., Fernandes, M.J. & Carvalho, J.B.D.C. (2015) "The international harmonization process of public sector accounting in Portugal: the perspective of different stakeholders", *International Journal of Public Administration*, vol. 38, no. 4: 268-281
- González, R.M.D., Julve, V.M. & Bargues, J.M.V. (2018). "Towards convergence of government financial statistics and accounting in Europe at central and local levels", *Revista de Contabilidad – Spanish Accounting Review*, vol. 21, no. 2: 140-149

Vol. 21, No. 4

- Heald, D. & Hodges, R. (2018) "Accounting for government guarantees: perspectives on fiscal transparency from four modes of accounting", *Accounting and Business Research*, vol. 48, no. 7: 782-804
- Heiling, J., Schührer, S. & Chan, J.L. (2013) "New development: towards a grand convergence? International proposals for aligning government budgets, accounts and finance statistics", *Public Money & Management*, vol. 33, no. 4: 297-303
- Hung, M. & Subramanyam, K.R. (2007) "Financial statement effects of adopting international accounting standards: the case of Germany", *Review of Accounting Studies*, vol. 12, no. 4: 623-657
- Hyndman, N. & Connolly, C. (2011) "Accruals accounting in the public sector: a road not always taken", *Management Accounting Research*, vol. 22, no. 1: 36-45
- International Public Sector Accounting Standards Board (IPSASB) (2014a) "Process for considering GFS reporting guidelines during development of IPSASs", available on-line at https://www.ifac.org/system/files/publications/ files/IPSASB-GFS-Policy-Paper.pdf (Retrieved November 15, 2022)
- International Public Sector Accounting Standards Board (IPSASB) (2014b) "The conceptual framework for general purpose financial reporting by public sector entities", available on-line at https://www.ifac.org/system/files/publications/files/IPSASB-Public-Sector-Conceptual-Framework.pdf (Retrieved November 15, 2022)
- Ismail, S. (2022) "Perception of the Malaysian Federal Government accountants of the usefulness of financial information under an accrual accounting system: a preliminary assessment", *Meditari Accountancy Research*, in press
- Jesus, M.A. & Jorge, S.M. (2014) "From governmental accounting into national accounts: adjustments diversity and materiality with evidence from the Iberian countries' central governments", *Innovar*, vol. 24, no. 54: 121-138
- Jorge, S., Vaz de Lima, D., Pontoppidan, C.A. & Dabbicco, G. (2019) "The role of charts of account in public sector accounting", *Paper presented at the II Congresso Internacional de Contabilidade Pública*, Lisbon, Portugal, available on-line at https://research-api.cbs.dk/ws/portalfiles/portal/58520266/ Jorge_VazdeLima_AggestamPontoppidan_Dabbicco.pdf (Retrieved November 15, 2022)
- Kadri, M.H., Abdul Aziz, R. & Ibrahim, M.K. (2009) "Value relevance of book value and earnings: evidence from two different financial reporting regimes", *Journal of Financial Reporting and Accounting*, vol. 7, no. 1: 1-16. doi:10.1108/19852510980000638
- Kober, R., Lee, J. & Ng, J. (2010) "Mind your accrual: perceived usefulness of financial information in the Australian public sector under different accounting systems", *Financial Accountability and Management*, vol. 26, no. 3: 267-298
- Kothari, S.P. & Zimmerman, J.L. (1995) "Price and return models", *Journal of* Accounting and Economics, vol. 20, no. 2: 155-192

- Lampropoulou, M. & Oikonomou, G. (2018) "Theoretical models of public administration and patterns of state reform in Greece", *International Review of Administrative Sciences*, vol. 84, no. 1: 101-121
- Lequiller, F. & Derek, B. (2007) Understanding National Accounts, Paris: OECD Publishing
- Lin, Z.J. & Chen, F. (2005) Value relevance of international accounting standards harmonization: evidence from A- and B-share markets in China", *Journal of International Accounting, Auditing and Taxation*, vol. 14, no. 2: 79-103
- Lorson, P.C., Jorge, S. & Haustein, E. (eds.) (2019) *European Public Sector Accounting*, Coimbra: Coimbra University Press
- Manes-Rossi, F., Cohen, S., Caperchione, E. & Brusca, I. (2016) "Harmonizing public sector accounting in Europe: thinking out of the box", *Public Money & Management*, vol. 36, no. 3, 189-196
- Moore, D.S., Notz, W.I. & Fligner, N.A. (2015) *The Basic Practice of Statistics*, 7th edition, New York: W. H. Freeman and Company
- Mussari, R. (2014) "EPSAS and the unification of public sector accounting across Europe", *Accounting, Economics and Law*, vol. 4, no. 3: 299-312
- Nitzl, C., Hilgers, D., Hirsch, B. & Lindermüller, D. (2020) "The influence of the organizational structure, environment, and resource provision on the use of accrual accounting in municipalities", *Schmalenbach Business Review*, vol. 72, no. 2: 271-298
- Ohlson, J.A. (1995) "Earnings, book values, and dividends in equity valuation", *Contemporary Accounting Research*, vol. 11, no. 2: 661-687
- Ohlson, J.A. (2001) "Earnings, book values, and dividends in equity valuation: an empirical perspective", *Contemporary Accounting Research*, vol. 18, no. 1: 107-120
- Organisation for Economic Co-operation and Development (OECD)/International Federation of Accountants (IFAC) (2017) Accrual Practices and Reform Experiences in OECD Countries, Paris: OECD Publishing
- Pontoppidan, C.A. & Brusca, I. (2016) "The first steps towards harmonizing public sector accounting for European Union member states: strategies and perspectives", *Public Money & Management*, vol. 36, no. 3: 181-188
- PricewaterhouseCoopers (PwC) (2014) "Collection of information related to the potential impact, including costs, of implementing accrual accounting in the public sector and technical analysis of the suitability of individual IPSAS standards. Report 2013/S 107-182395", available on-line at https://ec.europa.eu/eurostat/documents/1015035/4261806/EPSAS-study-final-PwC-report.pdf (Retrieved November 15, 2022)
- Salato, R.C.J., Gomes, P. & Ferreira, C. (2022) "Transition to accrual accounting in the public sector of emerging economies: a literature review", *International Journal of Business Innovation*, vol. 1, no. 1: e27646

Vol. 21, No. 4

- Srivastava, A. & Muharam, H. (2021) "Value relevance of accounting information during IFRS convergence period: comparative evidence between India and Indonesia", *Accounting Research Journal*, vol. 35, no. 2: 276-291
- Temiz, H. & Güleç, Ö.F. (2017) "Mandatory adoption of IFRS in emerging markets: the case of Turkey", *Accounting and Management Information Systems*, vol. 16, no. 4: 560-580
- Tsalavoutas, I., André, P. & Evans, L. (2009) "Transition to IFRS and value relevance in a small but developed market: a look at Greek evidence", available on-line at https://halshs.archives-ouvertes.fr/halshs-00460532/document (Retrieved November 15, 2022)
- Verrinder, J. (2021) "Current state of the EPSAS project and matters of sustainability", *Paper presented at the 18th Biennial CIGAR Conference*, Vrije Universiteit Amsterdam, The Netherlands, 24-25 June
- World Bank/Public Sector Accounting and Reporting Program (PULSAR) (2021) "Drivers of public sector accounting reforms", available on-line at https://cfrr.worldbank.org/sites/default/files/2021-02/pulsar_drivers_0.pdf (Retrieved November 15, 2022)
- Zaja, M.M., Jakovcevic, D. & Visic, L. (2018) "Determinants of the government bond yield: evidence from a highly Euroised small open economy", *International Journal of Economic Sciences*, vol. 7, no. 2: 87-106

Vol. 21, No. 4